RESEARCH

Journal of Eating Disorders

Open Access

Differences in sleep patterns between patients with anorexia nervosa and healthy controls: a cross-sectional study



Malin Mandelid Kleppe^{1,2*}, Ute Kessler^{1,3}, Guro Årdal Rekkedal^{4,5}, Hanna Flækøy Skjåkødegård^{6,7,8} and Yngvild Sørebø Danielsen^{1,4*}

Abstract

Background Sleep difficulties are common in patients with anorexia nervosa (AN), but objective assessments have mostly been performed in hospital and laboratory settings. We aimed to identify differences in sleep patterns between patients with AN and healthy controls (HC) in their free-living environments, and potential associations between sleep patterns and clinical symptoms in patients with AN.

Methods This cross-sectional study analyzed 20 patients with AN prior to them starting outpatient treatment and 23 HC. Sleep patterns were measured objectively using an accelerometer (Philips Actiwatch 2) for 7 consecutive days. Average sleep onset, sleep offset, total sleep time, sleep efficiency, wake after sleep onset (WASO) and mid-sleep awakenings lasting \geq 5 min were compared between patients with AN and HC using nonparametric statistical analyses. Associations of sleep patterns with body mass index, eating-disorder symptoms, eating-disorder-associated impairment, and symptoms of depression were assessed in the patient group.

Results Compared with HC, patients with AN had shorter WASO [median (interquartile range(IQR)): 33 vs. 42 min], but a longer average duration of mid-sleep awakenings lasting \geq 5 min [median (IQR): 9 vs. 6 min, p = 0.006] and had more nights with no sleep (six nights in four patients with AN vs. zero nights in HC). There were no differences between patients with AN and HC regarding other sleep parameters and no significant correlations between sleep patterns and clinical parameters in patients with AN. However, HC presented a Intraindividual variability pattern that was closer to a normal distribution, whereas patients with AN tended to either have very regular or large variability in sleep onset time (AN; n = 7 < 25th percentile and n = 8 > 75th percentile vs. HC; n = 4 < 25 percentile and n = 3 > 75th percentile) during the week of sleep recordings.

Conclusion Patients with AN seem to spend more time awake during the night and have more nights without sleep than do HC, even though their average weekly sleep duration did not differ from that in HC. The intraindividual variability in sleep pattern seems to be an important parameter that should be assessed when studying sleep in patients with AN.

Trial registration ClinicalTroals.gov. Identifier: NCT02745067. Registered: April 20, 2016.

Keywords Anorexia nervosa, Actiwatch 2, Sleep, Intraindividual variability

*Correspondence: Malin Mandelid Kleppe malin.mandelid.kleppe@helse-bergen.no Yngvild Sørebø Danielsen Yngvild.Danielsen@uib.no Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/A.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain English summary

Sleep disturbances are common in patients with anorexia nervosa. However, most studies that have assessed sleep patterns in this patient group have used either sleep questionnaires or sleep-measuring instruments in surroundings different from where they usually sleep. This study compared sleep patterns between patients with anorexia nervosa and healthy individuals without an eating disorder in their home environments using an accelerometer over 1 week. The accelerometer used was a sleep-monitoring instrument that looks similar to a wristwatch and records sleep patterns using movement and light sensors, and is easy to use in the home environment. We observed that patients with anorexia nervosa on average had longer periods awake during the night, as well as more awake nights compared with healthy individuals. Nevertheless, the average sleep duration during the week of sleep recordings was similar in the two groups. However, we also observed considerable variations in sleep patterns among the patients with anorexia nervosa, ranging from very regular to very irregular sleep patterns. We therefore believe that variability in the sleep patterns among patients with anorexia nervosa is an important parameter that should be included when studying sleep in these patients.

Background

Anorexia nervosa (AN) is a mental health disorder characterized by abnormal eating patterns, severe self-induced weight loss, intense fear of weight gain, and a disturbed body perception [1]. Patients with AN often report a lower quality of life and a range of serious somatic complications associated with the disorder [2]. The main treatment option is psychotherapy targeting factors that induce and maintain AN, such as cognitive behavioral therapy for

eating disorders or family-based treatment for AN [3]. However, patients often find these treatments challenging, and a large proportion of them leave their treatment program prematurely or suffer from reoccurring symptoms [4–6]. Some studies have found that reduced sleep quality and fragmented sleep can form part of the complex presentation of behaviors and complications of AN [7, 8]. Kim et al. assessed the prevalence of sleep disturbances by applying structured interviews to 400 patients with eating disorders, and found that 58% of patients with AN reported sleep disturbances, most commonly difficulty falling asleep and mid-sleep awakenings [7]. Patients undergoing treatment for AN may find that poor sleep increases the challenge of actively engaging in treatment, since a lower sleep quality has been associated with a higher risk of dropping out and with remission not occurring during treatment [9].

Sleep can be measured either subjectively or objectively using several different methods. Laboratory-based polysomnography (PSG) measures sleep objectively using electroencephalography (EEG), electromyography, and electro-oculography, and is considered the gold-standard assessment method for sleep architecture and associated disorders [10]. The results from early studies using EEG (the precursor to PSG) to compare sleep characteristics between patients with AN and healthy controls (HC) were inconsistent. Some studies found no or minimal difference in sleep patterns between patients with AN and HC [11, 12], while others found shorter total sleep time [13, 14], less stage 1 (light sleep) sleep [13], more stage 1 sleep, and less stage 3 sleep (slow-wave sleep) [14], in addition to increased sleep latency and more sleep disruption in patients with AN. A more recent study employing PSG revealed that relative to HC, patients with AN had longer periods of light sleep and shorter periods of slow-wave sleep, longer sleep latency, reduced sleep efficiency, and more arousals [8]. However, PSG measurements are resource-demanding and are most frequently conducted in an in-hospital setting, which might influence the measured behaviors [10]. In contrast, accelerometers are easily available, less invasive, and can still detect sleep patterns in free-living environments, and several types of accelerometers have been validated against PSG for use in sleep assessments [15, 16]. A study using accelerometers to assess sleep found that the total sleep time and sleep onset latency were both shorter in 50 patients recently admitted to inpatient care for AN than in HC [17]. However, the sleeping behaviors observed in inpatients might be markedly influenced by department routines. Most patients with AN receive outpatient treatment, which makes it important to obtain information about their sleep patterns in their home environments. To our knowledge, only one study has assessed sleep patterns in patients with AN in a natural environment using objective methods: Latzer et al. used accelerometers in 21 outpatients before starting treatment, and found no differences in sleep patterns between patients with AN and HC [18].

One of the possible causes of sleep problems in patients with AN is malnutrition and accompanying hormonal changes [19]. Moreover, depression is one of the most common comorbid disorders of AN [20], and is well known to impact sleep [21]. This situation makes it relevant to determine the extent to which symptoms of depression and the severity of underweight are related to sleep patterns in patients with AN. A better understanding of sleep patterns in AN is important because sleep disturbances in this group have been associated with worse treatment outcomes [9], and to more-severe eating-disorder psychopathology [7]. Moreover, the sleep patterns in patients with AN living in their home environments have not been extensively researched, and more knowledge about sleep patterns in this group would facilitate clinical decision-making about treatments.

The main aim of this study was to identify [1] differences in sleep patterns between patients with AN and HC in their home environments, and [2] potential associations between sleep patterns and the following clinical symptoms in patients with AN: body mass index (BMI), eating-disorder symptoms, psychosocial impairment due to eating disorders, and symptoms of depression.

Methods

This study formed part of a treatment trial that is described elsewhere [22], and had a cross-sectional case– control design. Patients were assessed prior to starting outpatient treatment. The diagnosis of AN (including atypical AN with BMI>18.5 kg/m²) was established by the treating psychologist according to criteria in the Diagnostic and Statistical Manual of Mental Disorders (fifth edition) [1].

Participants

All patients aged > 16 years who agreed to receive cognitive behavioral therapy for AN at the Department for Eating Disorders outpatient unit at Haukeland University Hospital, Bergen, Norway between December 2016 and August 2019 were asked to participate in the study. Thirty-three patients with AN agreed to participate by signing written informed-consent forms, among which 20 were eligible for inclusion in the statistical analysis; 12 of the excluded patients did not use the accelerometer measuring sleep, and 1 other was excluded due to having a BMI > 20 kg/m² (21.3 kg/m²). Twenty-three agematched HC with BMI values between 20 and 30 kg/m² were recruited through advertisements at the local university and college.

Anthropometric measures

The height and weight in patients with AN were measured by their therapist, while the study nurse measured the HC, with the values used to calculate BMI.

Measures of clinical symptoms

Eating disorder symptoms and the severity of psychosocial impairment associated with eating disorders were evaluated using the Eating Disorder Examination Questionnaire (EDE-Q global score) [23, 24] and the Clinical Impairment Assessment (CIA) [25], respectively. Symptoms of depression were evaluated using the Beck Depression Inventory-II (BDI-II) [26].

Sleepmeasures

Home environment sleep was measured using an accelerometer [Philips Actiwatch 2 (AW2)] placed on the participant's nondominant arm (wrist) for 7 consecutive days. The AW2 device measures sleep patterns using movement and light sensors [27], and has been validated as a sleep detection device in a comparison with PSG (the gold standard) [16]. A minimum of 5 days recordings were required for the participant to be included in the analysis. Data were collected using 30-second epochs, scored as awake/sleep based on a medium sensitivity threshold.

Sleep statistics were calculated using the Respironics Actiware software (version 6.0.9). A standardized scoring protocol [27] was used to define the rest interval (time in bed) at nighttime (rest intervals that started between 09:00 and 21:00 were not assessed in this study). The standardized scoring protocol took into account both the movement and light statuses when defining rest intervals (time in bed). All files were manually inspected by two scorers to ensure that rest intervals were scored in accordance with the protocol during the defined analysis time period. Sleep patterns during the defined rest intervals at night were further detected by a standard default algorithm in the Respironics Actiware software (version 6.0.9).

The sleep variables assessed included average sleep onset, and offset time, total sleep time (total duration of sleep during the major sleep period), sleep efficiency (proportion of time the participant is asleep during the total time in bed), waking up after sleep onset (WASO) and mid-sleep awakenings lasting ≥ 5 min (their frequency and average duration). The software automatically detects all periods of WASO according to movement measurements (including movement periods lasting only 30 s), and therefore we decided to also consider midsleep awakenings lasting ≥ 5 min (scored manually). We also assessed intraindividual variability (IIV) in the sleep onset time, sleep offset time, and total sleep time, by calculating the individual standard deviation (SD) (square root of the variance) across the nights of measured sleep in each participant [28].

Statistical analyses

Data were analyzed using SPSS software (version 26.0.0.1). The presence of a normal distribution was evaluated using the Kolmogorov-Smirnov test, as well as assessing skewness and kurtosis, and histogram and Q-Q plots. Nonparametric tests were applied since most of the sleep variables did not conform to a normal distribution. The Mann-Whitney *U*-test for independent samples was used to compare group medians. Data are presented as median and interguartile range (IQR) values, and the participant with the smallest and largest average values in each group are presented to indicate ranges. Effect sizes (Cohen's d) of 0.2, 0.5, and 0.8 were taken to indicate small, medium, and large effects, respectively [29]. IIV from the whole sample (patients with AN and HC) was divided into 25th, 50th, and 75th percentiles, and we present the frequency for each group in the different percentiles. In patients with AN, associations between sleep variables and clinical characteristics were assessed using bivariate correlation analyses, and quantified using the Spearman correlation coefficient (rho), with rho = 0.10-0.29, 0.30-0.49, and 0.50-1.00 indicating weak, moderate, and strong relationships, respectively [29].

Results

Demographic and clinical characteristics

All participants were female. Patients with AN had a median self-reported illness duration of 8.7 years (range 1–33 years), with seven classified as having the restrictive type and thirteen as having the binge-eating/purging type. The demographic and clinical characteristics of patients with AN and HC are presented in Table 1. In the patient group, 5 patients used antipsychotics, 3 used antihistamine or melatonin, and 11 used antidepressants, with none using benzodiazepines or Z-hypnotics. None of the HC used medications that could impact sleep.

Differences in sleep parameters between patients with AN and HC

There were no differences between patients with AN and HC regarding average sleep onset, sleep offset, total sleep time, sleep efficiency, or number of mid-sleep awakenings lasting \geq 5 min (Table 2). However, patients with AN showed a larger variability in sleep patterns than did HC, as shown by a larger IQR, more awake nights with no sleep (six nights in four patients with AN vs. zero nights in HC). Patients also had shorter WASO, but longer duration of mid-sleep awakenings lasting \geq 5 min, which

 Table 1
 Demographic and clinical characteristics of patients with AN and HC

	Patients with AN (n=20)		HC (n=23)		Mann-Whitney U-test		
	Median (IQR)	Range	Median (IQR)	Range	p	Cohen's d	
Age, years	19.5 (8.0)	16.0-49.0	19.0 (7.0)	16.0-49.0	0.826	0.03	
BMI, kg/m ²	16.7 (2.8)	13.7-19.5	22.9 (3.5)	19.5-29.0	< 0.001	0.85	
EDE-Q global score	4.59 (1.60)	1.73-5.70	0.36 (0.55)	0.03-1.60	< 0.001	0.85	
CIA total score	38 (11)	15–48	1 (3)	0–7	< 0.001	0.85	
BDI-II total score	38 (19)	15-60	4 (4)	0-14	< 0.001	0.85	

Boldface indicates significant differences

Table 2 Con	mparing week	ly average s	leep patterns	per night	between	patients with	ו AN and	H
-------------	--------------	--------------	---------------	-----------	---------	---------------	----------	---

	Patients with A	N (n=20)	HC (n = 23)	Mann-Whitney <i>U</i> -test		
	Median (IQR)	Range	Median (IQR)	Range	р	Cohen's d
	00:19 (2:07)	22:43-03:44	00:33 (1:38)	23:16-02:15	0.903	0.02
Sleep offset (hh:mm)	07:59 (2:05)	06:13-13:37	08:16 (1:16)	06:26-10:57	0.981	< 0.01
Total sleep time (min)	409 (68)	309–502	425 (64)	357-523	0.752	0.05
Sleep efficiency (%)	85 (6)	76–92	85 (7)	75–90	0.181	0.20
Wake after sleep onset (WASO, min)	33 (14)	10–58	42 (18)	27–69	0.006	0.14
Mid-sleep awakenings lasting≥5 min (frequency)	0.93 (1.00) ^a	0–2	0.50 (1.00) ^b	0–2	0.062	0.29
Mid-sleep awakenings lasting \geq 5 min (duration, minutes)	9 (32) ^a	5–174	6 (1) ^b	5–38	0.006	0.42

^a n=19, ^bn=21

Boldface indicates significant differences

remained significant after applying Bonferroni correction (p < 0.007).

A visual inspection of data suggested differences in the IIV (individual SD for each participant across nights of measured sleep) of the sleeping patterns, especially regarding sleep onset in patients compared with HC. As indicated in Table 3, HC presented a IIV pattern that was closer to a normal distribution, whereas patients with AN tended to either have very regular sleep onset or large variability of sleep onset during the week of sleep recordings.

Association between sleep parameters and clinical measures in patients with AN

We further assessed if sleep characteristics were associated with clinical symptoms within the patient group. The results obtained in correlational analyses are presented in Table 4. There were positive correlations between several variables [EDE-Q global score and mean duration of mid-sleep awakenings lasting ≥ 5 min (p=0.036), CIA total score and sleep onset (p=0.029) and sleep offset (p=0.048), and BDI-II total score and mean sleep duration (p=0.039)], while BMI seemed to be unrelated to sleep patterns. However, none of the correlations remained significant after applying Bonferroni correction (p<0.002).

Table 3 Differences in IIV between patients with AN and HC

IIV of sleep onset			IIV of sleep offset			IIV of total sleep time			
Percentile (hh:mm)	AN (n=20)	HC (n=23)	Percentile (hh:mm)	AN (n=23)	HC (n = 23)	Percentile (hh:mm)	AN (n=20)	HC (n = 23)	
25th (≤0:45)	7 (35%)	4 (17%)	25th (≤1:00)	8 (40%)	3 (13%)	25th (≤0:42)	4 (20%)	7 (30%)	
50th	5 (25%)	16 (70%)	50th	8 (40%)	13 (57%)	50th	9 (45%)	12 (52%)	
75th (≥1:38)	8 (40%)	3 (13%)	75th (≥1:47)	4 (20%)	7 (30%)	75th (≥1:33)	7 (35%)	4 (18%)	

Table 4 Relationships of sleep variables with clinical characteristics of patients with AN

	BMI (kg/m²)		EDE-Q global score		CIA total score		BDI-II total score	
	p	rho	p	rho	p	rho	p	rho
Sleep onset time	0.474	- 0.170	0.152	0.332	0.029	0.478*	0.875	0.038
Sleep offset time	0.108	- 0.370	0.051	0.442	0.048	0.448*	0.150	0.334
Total sleep time	0.103	- 0.375	0.622	- 0.117	0.667	- 0.102	0.039	0.466*
Sleep efficiency	0.867	- 0.040	0.729	- 0.083	0.140	- 0.342	0.370	- 0.212
Wake after sleep onset (WASO)	0.661	0.105	0.486	- 0.165	0.071	0.766	0.539	0.146
Mid-sleep awakenings last- ing≥5 min (frequency)	0.693	-0.094	0.661	0.105	0.324	0.232	0.076	0.406
Mid-sleep awakenings last- ing≥5 min (duration, minutes)	0.293	-0.254	0.036	0.484*	0.413	0.199	0.246	0.280

Boldface indicates significant correlations

Discussion

This cross-sectional study compared sleep patterns between patients with AN and HC in their home environments. We found no significant differences in average sleep onset, sleep offset, total sleep time, or sleep efficiency between patients with AN and HC. However, patients with AN had more nights with no sleep, shorter WASO, longer periods of mid-sleep awakenings than did HC, as well as larger variability in sleep patterns. We detected that some of the patients with AN had very regular sleeping patterns, while others showed large IIV regarding the timing of sleep onset, sleep offset, and total sleep time.

Consistent with Latzer et al. also reporting on objectively measurements of sleep using accelerometers in patients with AN in a natural environment [18], the current study did not find differences relative to HC in sleep duration, timing, or efficiency. However, the current study found that patients with AN had a longer duration of mid-sleep awakenings lasting ≥ 5 min and more awake nights with no sleep. Latzer et al. found the awake time during night to be similar to that in HC. This difference could be due to how the awake time was measured in the two studies: because WASO is considered a quite unreliable parameter in distinguishing between movements during sleep and awake time [16], we therefore chose to also include awake episodes that lasted for ≥ 5 min [30]. It is notable that Latzer et al. found that the patients with

AN reported significantly more mid-sleep awakenings than did HC, together with more difficulties falling asleep and more daytime sleepiness [18]. Kim et al. [7] used clinical interviews to identify sleep disorders in individuals with eating disorders and found that a high percentage of them experienced mid-sleep awakenings. These findings indicate that patients with AN subjectively experience mid-sleep awakenings as a problem.

Contrary to our study, El Ghoch et al. [17] found that sleep duration as measured by accelerometers were shorter in patients with AN than in HC. Those authors measured sleep in an inpatient setting among patients whose BMI values were lower than in those included in our study and that of Latzer et al., which could be related to the shorter sleep duration reported for that study. A classic study from the 1950, which would not be considered ethical to carry out today, found reduced sleep in healthy participants who had been subjected to malnutrition [31]. However, BMI was not correlated with any of the sleep parameters measured in our study. Our results should not be interpreted as conflicting with the finding that being significant underweight and malnourished might be associated with sleep problems.

Studies assessing sleep in patients with AN using subjective methods (e.g., questionnaires and interviews) have reported difficulties falling asleep [7, 9]. However, El Ghoch et al. even found that patients with AN had significantly shorter sleep onset delay than did HC [17]. On the other hand, a study measuring sleep using PSG (the gold standard) found that patients with eating disorders (AN and bulimia nervosa) had a longer sleep onset latency than did HC [8]. More research using objective measures in larger samples is necessary to draw definitive conclusions about whether patients with AN have more problems than HC in falling asleep.

A particularly interesting finding in our study was that psychosocial impairment related to the eating disorder tended to be associated with a greater shift in the circadian rhythm with later sleep onset and sleep offset. The CIA questionnaire examines the extent to which an eating disorder negatively influences thoughts, feelings, and behaviors toward oneself or others. We could speculate that high scores are indicative of negative ruminations that might influence sleep. Further, a shifted circadian rhythm with later sleep timing often impairs functioning in daily life with regards to school, work, and social life. Helping affected patients to maintain a sleep–awake cycle that is compatible with normal social functioning might be important for their well-being and recovery.

In our study, the median sleep durations (6 h and 49 min for patients with AN, and 7 h and 5 min for HC) were slightly shorter than the recommended range of 7-9 h [32] for patients with AN and at the lower end of

the recommended range for HC. The lack of reaching recommendations in the patient group could have been related to the longer duration of mid-sleep awakenings.

From clinical settings it is known that some patients with AN may exhibit compulsive training during the night that compromises their sleep patterns. Visual inspections of the actograms revealed that awake nights and longer awake periods during nighttime in the patients with AN might have been related to them being physically active. Unfortunately, the algorithms in the scoring program for the AW2 accelerometers used in the current study did not allow both sleep and physical activity to examined over a 24-hour period, and to our knowledge no other studies have used accelerometers to examine physical activity during nighttime in patients with AN. Studies examining sleep and physical activity in this patient group during 24-hour periods would therefore be of interest. The current study indicated that patients with AN might have more awake nights and variability in sleep patterns than HC, even though the weekly average sleep durations in these two groups do not differ. A higher SD indicates a larger IIV, and so patients with AN were more likely than HC to have either a very regular or irregular sleeping pattern. This topic would be interesting to investigate further, since it could be of significance to disturbances in the circadian rhythm. It should also be noted that a higher IIV is more common in younger individuals [33].

The most important limitation of this study was the small sample, which means that the findings must be interpreted with caution. Another limitation was that sleep patterns were detected only at nighttime, which made it more difficult to obtain an accurate picture of the sleep habits in individuals who sleep during daytime. Further, accelerometers may interpret normal movements during sleep as WASO, and WASO measured by the AW2 device can differ significantly from PSG measurements [16]; we therefore decided to also include mid-sleep awakenings lasting ≥ 5 min. Accelerometers are generally reliable, but they can result in errors when distinguishing between participants who are sleeping or doing something else involving only small movements (e.g., playing computer games) [10], which especially makes it difficult to assess sleep during daytime. One way of monitoring if participants are actually sleeping or simply sedentary is for them to write sleeping diaries in parallel or click on a button on the accelerometer when going to bed and arising from bed, but these tools were not used in the present study.

Conclusions

Patients with AN seem to experience longer durations of mid-sleep awakenings and awake nights than do HC. A relatively large IIV in sleep patterns was observed in the patient group, and this therefore needs to be assessed when studying sleep in patients with AN.

Abbreviations

AN	Anorexia Nervosa
HC	Healthy Control
PSG	Polysomnography
EEG	Electroencephalography
BMI	Body Mass Index
EDE-Q	Eating DisorderExamination Questionnaire
CIA	Clinical ImpairmentAssessment
BDI-II	Beck DepressionInventory-II
AW2	Actiwatch 2
WASO	Waking up AfterSleep Onset
IIV	IntraindividualVariability
SD	Standard Deviation
IQR	Interquartile Range

Acknowledgements

The authors are grateful to all participants and want to thank Charlotte Jevne for recruiting the patients.

Author contributions

UK, GÅR and YD planned the study, MMK contributed with acquisition of data, HFS contributed with scoring the accelerometer (Philips Actiwatch 2), MMK, UK, HFS and YSD analyzed the data. MMK has drafted the work, UK and YSD were major contributors in writing the manuscript. All authors were consulted in the writing of the paper, all authors read and approved the final manuscript.

Funding

Open access funding provided by University of Bergen. This study was supported by the grant from the Norwegian Competence Center for Sleep Disorders (SOVno), and from Division of Psychiatry at the Haukeland University Hospital and Helse Vest (Health authorities in Western Norway) Grant Number: 912177.

Availability of data materials

Data are available from the corresponding author upon reasonable written request.

Declarations

Ethics approval and consent to participate

The study was approved by the Regional Committee for Medical and Health Research Ethics, Western Norway (REK Vest 2015/2328). All patients provided written informed consent prior to participation.

Consent for publication

Not applicable.

Competing interests

All authors declare that they have no competing interest to disclose.

Author details

¹ Department of Psychiatry, Haukeland University Hospital, Bergen, Norway.
 ² Department of Medicine, Haukeland University Hospital, Bergen, Norway.
 ³ Department of Clinical Medicine, University of Bergen, Bergen, Norway.
 ⁴ Department of Clinical Psychology, University of Bergen, Bergen, Norway.
 ⁵ Child and Family support, Municipality of Bergen, Norway.
 ⁶ Chinical Science, University of Bergen, Norway.
 ⁷ Children and Youth Clinic, Haukeland University Hospital, Bergen, Norway.
 ⁸ Department of Physiotherapy, Haukeland University Hospital, Bergen, Norway.

Received: 25 April 2023 Accepted: 26 April 2023 Published online: 16 May 2023

References

- American Psychiatric A. Diagnostic and statistical manual of mental disorders: DSM-5. 5th ed. ed. Washington, D.C: American Psychiatric Association; 2013.
- Mehler PS, Brown C. Anorexia nervosa medical complications. J Eat disorders. 2015;3:11.
- National Institute for. Health and Care Excellence [NICE]. Eating disorders: recognition and treatment. (NC69). 2017.
- Dejong H, Broadbent H, Schmidt U. A systematic review of dropout from treatment in outpatients with anorexia nervosa. Int J Eat Disord. 2012;45(5):635–47.
- Kessler U, Kleppe MM, Rekkedal G, Rø Ø, Danielsen Y. Experiences when implementing enhanced cognitive behavioral therapy as a standard treatment for anorexia nervosa in outpatients at a public specialized eating-disorder treatment unit. J Eat disorders. 2022;10(1):15.
- Frostad S, Danielsen YS, Rekkedal G, Jevne C, Dalle Grave R, Rø Ø, et al. Implementation of enhanced cognitive behaviour therapy (CBT-E) for adults with anorexia nervosa in an outpatient eating-disorder unit at a public hospital. J Eat disorders. 2018;6:12.
- Kim KR, Jung YC, Shin MY, Namkoong K, Kim JK, Lee JH. Sleep disturbance in women with eating disorder: prevalence and clinical characteristics. Psychiatry Res. 2010;176(1):88–90.
- Asaad Abdou T, Esawy HI, Abdel Razek Mohamed G, Hussein Ahmed H, Elhabiby MM, Khalil SA, et al. Sleep profile in anorexia and bulimia nervosa female patients. Sleep Med. 2018;48:113–6.
- Sauchelli S, Jiménez-Murcia S, Sánchez I, Riesco N, Custal N, Fernández-García JC, et al. Orexin and sleep quality in anorexia nervosa: clinical relevance and influence on treatment outcome. Psychoneuroendocrinology. 2016;65:102–8.
- Van de Water AT, Holmes A, Hurley DA. Objective measurements of sleep for non-laboratory settings as alternatives to polysomnography–a systematic review. J Sleep Res. 2011;20(1 Pt 2):183–200.
- 11. Levy AB, Dixon KN, Schmidt H. REM and delta sleep in anorexia nervosa and bulimia. Psychiatry Res. 1987;20(3):189–97.
- Lauer C, Zulley J, Krieg JC, Riemann D, Berger M. EEG sleep and the cholinergic REM induction test in anorexic and bulimic patients. Psychiatry Res. 1988;26(2):171–81.
- Walsh BT, Goetz R, Roose SP, Fingeroth S, Glassman AH. EEG-monitored sleep in anorexia nervosa and bulimia. Biol Psychiatry. 1985;20(9):947–56.
- 14. Levy AB, Dixon KN, Schmidt H. Sleep architecture in anorexia nervosa and bulimia. Biol Psychiatry. 1988;23(1):99–101.
- Roane BM, Van Reen E, Hart CN, Wing R, Carskadon MA. Estimating sleep from multisensory armband measurements: validity and reliability in teens. J Sleep Res. 2015;24(6):714–21.
- Shin M, Swan P, Chow CM. The validity of Actiwatch2 and SenseWear armband compared against polysomnography at different ambient temperature conditions. Sleep Sci (Sao Paulo Brazil). 2015;8(1):9–15.
- El Ghoch M, Calugi S, Bernabè J, Pellegrini M, Milanese C, Chignola E, et al. Sleep patterns before and after weight restoration in females with Anorexia Nervosa: a longitudinal controlled study. Eur Eat disorders review: J Eat Disorders Association. 2016;24(5):425–9.
- Latzer Y, Tzischinsky O, Epstein R. Sleep-wake monitoring in women suffering from anorexia nervosa. Eat Disord. 2001;9(2):159–66.
- 19. Lauer CJ, Krieg JC. Sleep in eating disorders. Sleep Med Rev. 2004;8(2):109–18.
- 20. Marucci S, Ragione LD, De Iaco G, Mococci T, Vicini M, Guastamacchia E et al. Anorexia Nervosa and Comorbid Psychopathology. Endocrine, metabolic & immune disorders drug targets. 2018;18(4):316–24.
- Allison KC, Spaeth A, Hopkins CM. Sleep and eating Disorders. Curr psychiatry Rep. 2016;18(10):92.
- Danielsen YS, Årdal Rekkedal G, Frostad S, Kessler U. Effectiveness of enhanced cognitive behavioral therapy (CBT-E) in the treatment of anorexia nervosa: a prospective multidisciplinary study. BMC Psychiatry. 2016;16(1):342.

- Fairburn CGB. Eating disorder examination questionnaire. In: Fairburn CG, editor. Cognitive behavioral therapy and eating disorders. New York, NY: Guilford Press; 2008. pp. 309–13.
- Rø Ø, Reas DL, Stedal K. Eating disorder examination Questionnaire (EDE-Q) in norwegian adults: discrimination between female controls and eating disorder patients. Eur Eat Disord Rev. 2015;23(5):408–12.
- Bohn K, Doll HA, Cooper Z, O'Connor M, Palmer RL, Fairburn CG. The measurement of impairment due to eating disorder psychopathology. Behav Res Ther. 2008;46(10):1105–10.
- Beck AT, Steer RA, Brown GK. Beck depression inventory (BDI-II):Pearson; 1996.
- Chow CM, Wong SN, Shin M, Maddox RG, Feilds KL, Paxton K, et al. Defining the rest interval associated with the main sleep period in actigraph scoring. Nat Sci sleep. 2016;8:321–8.
- Bredeli E, Vestergaard CL, Sivertsen B, Kallestad H, Øverland S, Ritterband LM, et al. Intraindividual variability in sleep among people with insomnia and its relationship with sleep, health and lifestyle factors: an exploratory study. Sleep Med. 2022;89:132–40.
- Cohen J. Statistical power analysis for the behavioral Sciences. 2nd ed. Hillsdale, NJ: Lawrence Erlbaum Associates, Publishers; 1988.
- Langlet B, Vestermark F, Stolt J, Zandian M, Södersten P, Bergh C. Physical activity and sleep during the first week of anorexia nervosa inpatient care. PLoS ONE. 2021;16(11):e0260077.
- Keys A. The residues of malnutrition and starvation. Sci (New York NY). 1950;112(2909):371–3.
- National Center for Chronic Disease Prevention and Health Promotion DoPH. How Much Sleep Do I Need? [updated Last Reviewed: September 14, 202211.12.22]. Available from: https://www.cdc.gov/sleep/about_ sleep/how_much_sleep.html.
- Bei B, Wiley JF, Trinder J, Manber R. Beyond the mean: a systematic review on the correlates of daily intraindividual variability of sleep/wake patterns. Sleep Med Rev. 2016;28:108–24.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

